This is the peer reviewd version of the followng article:

Renewable energy in eco-industrial parks and urban-industrial symbiosis: A literature review and a conceptual synthesis / Butturi, M. A.; Lolli, F.; Sellitto, M. A.; Balugani, E.; Gamberini, R.; Rimini, B.. - In: APPLIED ENERGY. - ISSN 0306-2619. - 255:(2019), pp. 1-16. [10.1016/j.apenergy.2019.113825]

Terms of use:

The terms and conditions for the reuse of this version of the manuscript are specified in the publishing policy. For all terms of use and more information see the publisher's website.

06/08/2024 06:39

# Renewable energy in eco-industrial parks and urban-industrial symbiosis: a literature review and a conceptual synthesis

DOI: 10.1016/j.apenergy.2019.113825

Link: https://linkinghub.elsevier.com/retrieve/pii/S0306261919315120

Butturi, M.A.<sup>1,\*</sup>, Lolli, F.<sup>1,2</sup>, Sellitto, M.A.<sup>3</sup>, Balugani, E.<sup>1</sup>, Gamberini, R.<sup>1,2</sup>, Rimini, B.<sup>1,2</sup>

<sup>1</sup> Department of Sciences and Methods for Engineering (DISMI), University of Modena and Reggio Emilia, Via Amendola 2 – Padiglione Morselli, 42100 Reggio Emilia, Italy

<sup>2</sup> EN&TECH Interdepartmental Center (University of Modena and Reggio Emilia), Piazzale Europa 1, 42100 Reggio Emilia, Italy

<sup>3</sup> Production and Systems Engineering Graduate Program, Universidade do Vale do Rio dos Sinos -Unisinos, Av. Unisinos 950 São Leopoldo 93.022-000, Brazil

#### Abstract

Replacing fossil fuels with renewable energy sources is considered as an effective means to reduce carbon emissions at the industrial level and it is often supported by local authorities. However, individual firms still encounter technical and financial barriers that hinder the installation of renewables. The eco-industrial park approach aims to create synergies among firms thereby enabling them to share and efficiently use natural and economic resources. It also provides a suitable model to encourage the use of renewable energy sources in the industry sector. Synergies among eco-industrial parks and the adjacent urban areas can lead to the development of optimized energy production plants, so that the excess energy is available to cover some of the energy demands of nearby towns. This study thus provides an overview of the scientific literature on energy synergies within eco-industrial parks, which facilitate the uptake of renewable energy sources at the industrial level, potentially creating urban-industrial energy symbiosis. The literature analysis was conducted by arranging the energy-related content into thematic categories, aimed at exploring energy symbiosis options within eco-industrial parks. It focuses on the urban-industrial energy symbiosis solutions, in terms of design and optimization models, technologies used and organizational strategies. The study highlights four main pathways to implement energy synergies, and demonstrates viable solutions to improve renewable energy sources uptake at the industrial level. A number of research gaps are also identified, revealing that the energy symbiosis networks between industrial and urban areas integrating renewable energy systems, are under-investigated.

#### <u>Keywords</u>

Carbon emissions reduction; Renewable energy sources; Industrial symbiosis; Eco-industrial parks; Industrial energy symbiosis; Urban-industrial energy symbiosis

#### Highlights

- Options to reduce industry GHG emissions
- Review and analysis of energy symbiosis schemes including renewable energy sources
- Energy strategy within eco-industrial parks to promote the use of renewable energy sources
- Urban-industrial energy symbiosis including renewable energy sources

#### 1. Introduction

<sup>\*</sup> Corresponding author. E-mail address: <u>mariaangela.butturi@unimore.it</u> (Maria Angela Butturi)

Reducing the negative environmental impacts of industries is a major challenge, both in advanced and emerging economies. The need for reducing atmospheric emissions, which are responsible for climate change, is all the more urgent in densely industrialized areas, which are traditionally located nearby highly populated zones, such as those in Europe [1]. The quality of air in rapidly growing economies is also worsening, as industrial clusters bring about both economic growth and environmental impacts [2]. Issues of environmental resource efficiency, sustainable processes and energy choices must be considered when promoting sustainable industrial development [3].

Global CO<sub>2</sub> emissions from the industry sector include energy-related emissions and CO<sub>2</sub> emissions from industrial processes. Direct emissions mainly originate from fuel combustion and industrial processes, while indirect emissions are primarily from energy production (electricity and heat) [4]. Therefore decarbonization processes should be based on the sustainable transformation of production processes and sustainable energy sources [5].

The Industrial Development Report 2018 of the United Nations Industrial Development Organization [6] reaffirms that industries should create a "virtuous circle of sustainable consumption is a system in which fossil fuel inputs are gradually replaced with renewable energy, materials and energy are used more efficiently, and final goods are reused or recycled to feed back into the input-generation process".

Improvements in energy and material efficiency, and a greater deployment of renewable energy, are considered as essential for a low-carbon transition [7]. The potential for CO<sub>2</sub> emission reduction offered by renewable energy sources (RES) in energy production and industrial processes is emphasized by the International Energy Agency [8]. Industries can buy clean energy (e.g., hydropower) for processes or integrate renewable energy systems. Technological advancements now provide solutions for adopting intermittent RES through methods such as distributed energy generation [9] and smart grids [10]. The main renewable energy sources suitable for industrial application are biomass, solar radiation (thermal or photovoltaics), ground heat and wind. The maturity and cost reduction of renewables technology, in which renewable resources are abundant and the cost of energy produced by RES is getting closer to fossil fuel parity, makes RES a feasible alternative.

However, single firms, and particularly single small-medium enterprises (SMEs), are reluctant to invest in energy efficiency solutions and RES, mainly due to techno-economic factors [11]. Decision-making in energy projects requires expertise, which is often not internally available, and involves technical, economic, environmental and social issues that SMEs may find difficult to manage [12].

Supported supply chain initiatives have been demonstrated to be effective in improving energy efficiency, leading to greenhouse gas (GHG) emission reductions and cost savings both for

suppliers and manufacturers [13]. Approaching sustainable solutions and energy planning from the district level can encourage the implementation of inter-firm projects that promote energy exchanges and collective production to reduce the use of fossil fuels. The sharing of the same geographical and administrative conditions (climate, energy stakeholders, local policies and networks) represents an opportunity to implement energy strategies aimed at rationalizing consumption and optimizing the systems of supply [14].

Eco-industrial parks (EIPs) are naturally suited to foster cooperation and resource-sharing among businesses. EIPs comprise a community of businesses located in the same geographical area, connected by collaborative [15] and competitive relationships [16]. EIPs can be considered an application of the industrial symbiosis (IS) approach to industrial systems. IS rests on building synergies among different firms to develop closed-loop systems through the physical exchange of resources (materials, by-products, water, energy), thus reducing pollution and the exploitation of natural resources. EIPs are modelled as networks of manufacturing and service businesses, which aim to share and efficiently use natural and economic resources, increase the economic performance of the participants, reduce the overall environmental impact and create benefits for local communities [17].

Thus, a collective effort to increase the use of RES is a viable method of reducing the carbon footprint of industry as cooperation among businesses enables knowledge gaps to be overcome and the investment, maintenance, and management costs of the energy infrastructure to be shared. The industrial sites can evolve into energy producers, able to satisfy internal energy demands and also to supply neighbouring populated areas with the excess energy [18], thus minimizing the environmental impact of electricity production [19]. Urban-industrial symbiosis (UIS) extends the concept of industrial symbiosis to urban-industrial synergies, and provides the opportunity of building up fruitful exchanges of resources and waste between cities and local EIPs, and reducing carbon emissions both at industrial park and urban levels [20].

Research into EIPs is well-established and the published literature provides a consolidated body of knowledge. Reviews of EIP studies [21–23] and of RES in industrial applications [24–26] have been presented. However, to the best of our knowledge no review integrates these issues or explores the possibilities of EIP and RES integration, which combine the industrial and the nearby urban levels. Thus, the aim of this study is to fill this research gap and investigate how EIP models can help promote the use of RES, at both industrial and urban levels.

This paper presents an overview of the scientific literature on energy synergies within EIPs that enable RES uptake at the industrial level. It provides a framework for the knowledge on energy sustainability in industrial parks and outlines the inter-firm cooperation schemes that can foster the adoption of emission-mitigating technologies. It demonstrates viable solutions and the effectiveness of RES integration within EIPs aimed at reducing the carbon footprint of industries, and potentially creating urban-industrial energy symbiosis. Finally, it suggests opportunities for future research.

The remainder of this article is structured as follows. Section 2 presents the research motivation and questions. Section 3 describes the literature search, including keywords used and the subsequent steps. Section 4 provides an analysis of the selected articles, arranged by thematic content. Section 5 includes a discussion of the findings and a conceptual synthesis highlighting the issues and possibilities for future research. Lastly, section 6 draws the conclusions.

# 2. Motivation and research questions

Many literature reviews of research into issues related to eco-industrial parks have been found, which analyse both academic areas of interest and the features of projects, as for example in [27]. The initial literature search was conducted using the Science Direct and Web of Science databases, combining the terms "industrial symbiosis", "eco-industrial parks" and "energy" and selecting papers classified as "reviews". Table 1 provides a summary of relevant recently published surveys or reviews (11 out of 25) on the topic. The table also reports the number of articles that each study reviewed, used for a backward search.

Focus	Keywords	N. of reviewed	Year	Source
		papers		
Policy conceptualization	Industrial symbiosis;	37 out of 118	2014	[21]
to clarify the	Policy intervention;	references		
mechanisms of policy	Policy translation;			
intervention and	Circular economy;			
facilitation of industrial	Eco-industrial park;			
symbiosis.	China			
Industrial ecology	Circular Economy;	43 out of 110	2018	[28]
contribution to Circular	Industrial Symbiosis:	references		
economy				
	Industrial Ecology;			
	Bibliometric analysis			
	Dibilometric analysis			
Optimization methods for	Eco-industrial parks;	44 references	2015	[22]
designing eco industrial	Industrial ecology;			
parks	Optimization;			
	Energy; Water			
	network;			

	Mathematical			
	programming			
Eco-industrial parks in	Eco-Industrial Park;	55 references	2016	[29]
Korea	Eco-industrial			
	development;			
	Industrial symbiosis;			
	Korea;			
	Industrial			
	development policy			
Quantitative tools and	Industrial ecology	207 references	2015	[23]
methods to minimize	cultivation; Energy			
energy and material	network; Energy use			
consumption in IS	minimization; Water			
exchanges within	network; Retrofitting			
existing EIPs	industrial park; Eco-			
	industrial park			
Industrial symbiosis and	Eco-industrial park;	114 references	2011	[30]
energy strategies within	Mixed industrial			
EIPs	park; Energy			
	management;			
	Carbon footprint			
Sustainability transition	Eco-industrial parks;	66 case	2019	[31]
of industrial parks (EIP	Industrial parks;	studies out of		
development)	Sustainability	115 references		
	transitions; Strategic			
	niche management;			
	Systematic literature			
	review; Case survey			
Academic research on	Circular economy;	565 references	2018	[32]
Circular Economy	Sustainability;			
	Industrial ecology;			
	Systematic literature			
	review; Circular			
	business model			

Supply Chain Configurations in the Circular Economy	Circular economy; literature review; sustainable supply chains; closed-loop supply chains; industrial ecology	77 out of 452 references	2017	[33]
Circular economy	Industrial ecology;	151 references	2017	[34]
concept and applications	Industrial symbiosis;			
	Closed-loop;			
	Material flow; Waste			
	management;			
	Resource use			
Sustainability indicators	Sustainability	68 references	2016	[35]
for EIPs	indicators; Eco-			
	industrial parks;			
	Sustainability			
	assessment			

**Table 1.** List of relevant literature reviews combining the terms "industrial symbiosis", "ecoindustrial parks" and "energy", as returned by the online search. Where the number of reviewed papers is not explicitly provided, the number of references is reported.

As Table 1 indicates, few reviews focus on the energy theme. The development of EIP is analysed in relation to circular economy principles, the industrial ecology paradigm, and the industrial symbiosis approach. Technical, economic, environmental, and social impacts in addition to policy and sustainability implications are taken into consideration. An overview of energy management solutions to reduce the industry related carbon footprint within EIPs was presented by Maes et al. in 2011 [30], with the aim of designing an improved carbon neutrality strategy for industrial parks located in the Flanders Region of Belgium.

Since our preliminary research suggests that no recent literature review has assessed the possibilities of integrating EIP and RES, this paper aims to provide a comprehensive overview of the academic studies focusing on the energy strategies adopted in EIPs. The overview analyses the use of RES and possible urban-industrial synergies in order to identify the research gaps and to design a future research agenda.

Three main factors emerge from the research. Firstly, in terms of the influence of EIP models on energy concerns, the influence of IS approaches in energy management requires further investigation. Secondly, RES are found to be relevant in EIPs. Thirdly the energy management of

EIP has an impact on the neighbouring urban areas. In terms of these three factors, the literature analysis addresses the following questions, illustrated in Fig. 1:

- What types, if any, of industrial energy symbiosis have been devised, planned or implemented?
- To what extent are renewable energy sources used?
- How is urban-industrial energy symbiosis promoted as an integrated sustainable solution involving renewable energy sources?



# Fig. 1. The research questions.

The illustration of the research questions in Fig. 1 represents the industrial symbiosis (grey arrows) and the investigated potential energy symbiosis (red-dotted arrows) within the EIP. Moreover, the contribution of RES is investigated, supporting energy synergies among firms and urban-industrial symbiosis as well.

# 3. Review methodology

From the above literature review, we defined keywords for online searches in academic journal articles. The search was conducted using the online databases Science Direct and Web of Science.

In the first step, a list of relevant terms was identified, based on the focal topics: "Eco-industrial parks", "carbon emissions reduction", "industrial energy symbiosis", "renewable energy sources", and "urban-industrial energy symbiosis". We defined temporal boundaries for the review: the literature analysis was limited to articles published after 2010, when technological advancements and declining RES prices led to many techno-economic solutions for the integration of a high proportion of renewables in the energy system [36,37].

After a first exploration of the literature, a broad range of keywords and similar concepts emerged from our initial search terms. Table 2 presents an overview of the concepts associated with our research. The table combines relevant terms through an iterative process of refining the searches.

Initial terms	Relevant associated keywords and
	concepts
Eco-industrial parks	Industrial symbiosis
	Circular economy
	Industrial ecology
	Ecological industry chain
	Sustainability
	Eco-efficiency
	Industrial districts
	Industrial clusters
	Industrial synergies
Carbon emissions reduction	Greenhouse gas emissions reduction
	Low-carbon transition
	Carbon footprint
	Lifecycle assessment
Industrial energy symbiosis	Energy efficiency
	Energy savings
	Energy integration
	Inter-firm energy
	Energy clustering/clusters
Renewable energy sources	Solar renewable energy
	Multi-generation systems
	Multi-energy systems
	Smart grids

	Energy hub
	Distributed energy resources
	Storage
	Community
Urban-industrial energy symbiosis	Urban-industrial synergies
	District heating
	Urban energy system transition
	Eco-town
	Energy hub
	Regional multi-energy
	Community

**Table 2.** Concepts associated with the initial identified strings of terms that came to light after the first search of the scientific databases.

Using the titles and abstracts, we initially selected relevant papers and excluded those that clearly did not address our research. We then read the full texts of the remaining 291 articles and conducted a snowball search to ensure our research was as comprehensive as possible. A total of 88 papers were then selected as the most relevant. The yearly distribution of the selected papers is presented in Fig. 2.



Fig. 2. Yearly distribution of the selected papers.

As illustrated in Fig. 2, the most recent papers were selected to present the most up-to-date research. A few papers outside the set temporal boundaries were included in the review ([38], [39], [40], [41]) due to their seminal relevance to the topic under study.

## 4. Literature review results - Content thematic analysis

This section presents the results of the performed literature review, arranging the analysed content into thematic categories according to the research questions. The thematic analysis follows the outline illustrated in Fig. 3.



Fig. 3. Thematic outline of the reviewed literature.

Fig. 3 shows the three main constructs that emerged from the literature review: sustainable industrial sites, energy symbiosis within EIP, and urban-industrial energy symbiosis. The sustainability goals set within EIPs drive the creation of synergies among firms. These synergies can take the form of energy symbiosis and promote energy exchanges with local urban areas. The figure also shows a second construct derived from the papers analysis, resulting in energy symbiosis modelling, energy symbiosis involving RES in EIPs, and energy organizational strategies within EIPs. This is the structural axis for the review.

4.1 Sustainable industrial sites: the eco-industrial parks

The need to improve the sustainability of industrial sites is widely explored in the literature, mainly with the aim of developing solutions for reducing the environmental impact at the local level and supporting stakeholders and policy makers. The concept of industrial symbiosis, which has evolved through the academic debate [42–44] and specific projects, represents an effective approach to sustainability in the industry sector. It is considered a part of the industrial ecology field of knowledge [43] as it focuses on the optimization of the materials cycle and fulfils the circular economy principles of reusing, recycling and remanufacturing materials thereby increasing resource efficiency, reducing waste and pollution, and bringing about economic benefits [45].

According to the IS approach, eco-industrial parks are industrial clusters, i.e., a set of industrial companies located near each other, which take advantage of their geographical proximity and consider environment preservation as a key issue when developing cooperation among businesses, thus resulting in economic and social benefits. The network of firms is built on the sharing of resources that include materials, energy, water, infrastructures, and facilities, but also the natural habitat and information, and the exchanges of materials and energy aimed at minimizing the use of resources and the waste generated. EIP projects can be developed from spontaneous cooperation initiatives carried out by companies motivated to improve efficiency and cut costs (bottom-up-model) or promoted by governmental or other institutional initiatives (top-down model) [15]. Lambert and Boons [46] distinguish geographically concentrated industrial activities (mainly process activities with close physical couplings of a relatively small number of materials and energy intensive production processes), mixed industrial parks (where SMEs of different sectors with little coupling of production processes are concentrated in dedicated areas), and eco-industrial regions (referred to as administrative areas where diverse or related industrial enterprises are located).

Techno-economic factors affect the feasibility of EIPs and determine the possibility of creating networking relationships and exchanges among participants, in addition to organizational aspects such as the willingness of individual actors to participate in initiatives [22].

In an integrated approach to sustainability, the EIP is part of a wider system that include the local territory: viable EIPs are also characterized by beneficial interactions with the local community, considering both the social and economic dimensions of sustainable development [17]. Simeoni et al. [47] identify three progressive geographic boundary levels of intervention: endogenous, which refers to solutions designed to improve the sustainability of single activities; exogenous, which refers to solutions designed for industrial zones or parks; and industrial urban systems, which aim to integrate industrial parks into the neighbouring urban territory.

EIPs can be considered as innovation platforms facilitating action concerning environmental impacts, not only with the end-of-pipe approach but also at a systemic level, such as adopting a life-cycle perspective [34]. EIP sustainability according to this view is mainly evaluated in the

literature using life-cycle assessment (LCA) based analysis [48,49], mainly focusing on resources and GHG emissions. With reference to the eco-efficiency concept, as established in [50], the data envelopment analysis (DEA) method demonstrated its efficacy and is widely used to assess the sustainability of industrial parks. The relative eco-efficiency of industrial parks in China was analyzed by Fan et al. [51] who compared the factors influencing the eco-efficiency. Pai et al. [52] designed an eco-efficiency measurement model for EIPs in Taiwan, as a tool to be used for the continuous improvement of sustainability. Wang et al. [53] developed a methodology for the optimal design of the ecological industrial chain within EIPs based on the DEA method. Other methods, recently applied to evaluate the environmental efficiency and the GHG emissions of industrial processes and parks, are based on emergy analyses, considering the quality of each form of energy flowing through the system [54].

The sustainability objectives of EIP projects has been extensively studied, beginning with the Kalundborg industrial district in Denmark, which is considered a forerunner of the eco-industrial parks, because of the inter-firm and urban links initially developed to take advantage of an excess of energy [55]. Heeres et al. [56] investigate successful implementations in the US and the Netherlands. Chertow and Ehrenfeld [42] analyse a number of existing EIPs to derive a model for self-organizing systems.

Many recent papers study the results of various national programs supporting EIPs development: Park J. et al. [29] investigate the sustainability progress achieved by the first phase of the Korean EIP program, Huang et al. [57] review China's EIP system to provide tools for assessing EIPs, and Pilouk and Koottatep [58] study the Thai EIPs projects for developing a set of sustainability indicators suitable to promote EIPs. Domenech et al. [45] reported an overall updated overview of IS activity in Europe in their review, which was supported by the European Commission's funded research projects.

Energy, that is closely related to environment and sustainable growth, is a key theme when assessing industry sustainability objectives. Valenzuela-Venegas et al. [35] review a set of indicators for assessing the sustainability of eco-industrial parks in terms of energy. They list at least 20 possible indicators directly related to energy, many of them considered from the environmental point of view. To clarify the focus of these energy-related indicators, they have been categorized in Table 3 as energy efficiency, emission-related, energy consumption-related and resources use-related indicators (including energy).

Category	Indicator	Definition	Sustainability
			dimension
Energy efficiency	Energy consumption per	The energy efficiency of the candidate enterprise by calculating of all the	En

related indicators	unit	energy and converting to the number	
		of standard coal using means	
		conversion coefficients	
	Output rate of	The amount of production value in EIP	En/Ec
	energy	generated from one unit of energy	
	Energy	The level of efficient use of energy in a	En
	consumption per	firm	
	unit of		
	production value		
	Energy	The level of efficient use of energy in	En
	consumption per	the key industrial sector	
	unit of		
	production in the		
	key industrial		
	sector		
	Energy intensity	The energy consumption efficiency. It	En
		relates the consumption to the output	
		of the sector in monetary values	
	TEIw	The TEI per number of workers	En
	Direct energy	This refers to emissions from the direct	En
	consumption	combustion of fossil fuels within the	
	carbon footprint	administrative boundary	
	Electricity and	This refers to the indirect carbon	En
	heat carbon	footprint in terms of purchased	
Indicators	footprint	electricity and heat purchased out of	
		the park	
	Specific	The total CO <sub>2</sub> emissions related to the	En
	emission	energy consumption	
	Energy carrying	The possibility of meeting the energy	En
	capacity	demand of the candidate enterprise in	
Energy consumption		an EIP	
	Percent-added	The growth rate of energy production	En

	of park energy	in the park after the introduction of a	
	productivity	new business	
	Energy	The energy consumption including	En/Ec
	consumption per	coal, electricity, oil, and energy	
	added industrial	consumption for both heating and	
	value	cooling	
	Total energy	Sources of energy such as coal,	En
	consumption	electricity, oil and other energy	
	intensity	consumption (including the production	
		of heating and cooling energy) used for	
		the production and operation of the	
		enterprise	
	TEI	The amount of energy consumed by	En
		the system and subsystem,	
		differentiating between energy	
		generated domestically and energy	
		imported	
	Energy	The total energy consumption of a park	En
	consumption		
	indicator		
	Energy intensity	The sum of the total amount of energy	En
	Primary energy	The contribution of a material of a	En
		process to the primary energy	
	Resource use	This considers the three main	En
		resources of water, land, and energy	
Posourcos uso	Resource use	This is based on the overall resources	En
(including energy)	efficiency	including energy sources	
	Renewable	The total energy and material driving a	En
	resources input	process that is derived from renewable	
		sources	

**Table 3.** The main energy related indicators used in the literature for the sustainability assessment of EIPs (adapted from [35]. (En = Environmental; Ec = Economic)

Table 3 shows that a sustainability indicator considering the share of renewable energy and/or use within a park is not explicitly defined in the reviewed papers. Wang et al. [59] suggest an index system to evaluate the stability of the EIP network from the very beginning of the project including the renewable energy use ratio, i.e., the proportion of energy produced by renewable sources used by companies. This index considered an indicator of energy sustainability, is applied to assess the admissibility of a company to an EIP. According to this view, the new Chinese EIP standard system (set in 2015) introduces a new sustainability indicator "Usage rate of renewable resources" [57], as well as the Vietnamese programme that uses the share of energy produced by renewables as a key indicator for the EIP sustainability evaluation [60].

#### 4.2 Energy symbiosis within EIPs

In terms of industrial symbiosis, the electrical and thermal energy flows and the material flows among firms are designed with the aim of reducing the overall energy use and improving the air quality. Unlike material exchanges, which are specifically related to the industry sector of the involved enterprises, the energy symbiosis within EIPs can be approached in a more general way to develop technical and organizational strategies that can be compatible with mixed industrial parks and can comply with new or existing parks.

The energy use within an industrial site can be assessed detailing the activities conducted as industrial use (production-related equipment, including service facilities), building services use (utilities such as lightning, heating and cooling, safety systems, and transportation systems) and civil use (office buildings) [61]. The analysis of the thermal and electrical needs of companies and the comprehensive evaluation of energy inputs-outputs for all industrial processes enables an energy baseline to be calculated and defines where inter-plant or inter-company connections can be established.

In addition to inter-firm energy exchanges, joint projects for energy efficiency and for collective energy production can be implemented. Since the work of Fichtner et al. [38], different inter-firm energy supply concepts have been investigated as promising approaches for achieving cleaner energy production. In addition, EIPs are often already equipped with energy infrastructure for the transformation of waste materials into heat and electricity, which can be supplied to the enterprises joining the park or uploaded to the local power grid.

The energy symbiosis in EIPs is further investigated below according to the three constructs: the modelling approaches, the use of RES and the organizational strategies. This section answers the first two research questions, as represented in Fig. 4: the energy symbiosis within EIPs are investigated and the contribution of the RES is highlighted.



**Fig. 4.** The main renewable energy conversion technologies integrated within EIPs: waste/biomass treatment plants, wind turbines, solar panels (thermal collectors and photovoltaic plants), ground-source heat pumps. The carbon capture and storage (CCS) technology, used to prevent carbon emissions in the atmosphere is also depicted.

# 4.2.1 Energy symbiosis modelling

From a modelling point of view, the design of EIPs can be treated as a multi-objective problem because it involves numerous stakeholders with potentially conflicting objectives and a mix of technical, environmental and social issues. Techno-economic and environmental issues are the most addressed, while organizational and social aspects have been only recently considered [22].

Energy symbiosis modelling can be viewed as an aspect of EIP design. Most optimization methods used in the design of EIPs separately consider the types of symbiotic relationships, i.e., materials, energy, and water [22]. Some studies investigate interplant energy flow management, mainly using pinch analysis or mathematical programming approaches, and only a few address energy management in an EIP through mathematical optimization. Kastner et al. [23] reviewed modelling methods to identify and establish viable inter-company exchanges; pinch analysis and mixed integer linear programming are the main methods used to optimize energy (mainly heat) exchange networks.

Energy symbiosis models mainly aim to simultaneously minimize costs and emissions related to energy exchanges [62], and to optimize energy efficiency. Technical features should be taken into account, and both EIP advantages and the individual benefits should be considered [63]. Timmerman et al. [64] first compare energy models and then discuss a classification for them, to design a low carbon energy system within EIPs, suggesting a holistic techno-economic modelling approach. A low carbon energy system includes energy efficient technologies, maximizes the integration of local renewable energy sources and enables heat exchange between firms. Some of the features that a suitable energy system model should include are:

- multi-objective optimization, to facilitate the trade-off between conflicting objectives, such as minimisation of both costs and carbon emissions;
- a generic technology description at unit level;
- sufficient temporal detail, showing energy demand and RES availability trends and peaks
- energy storage technologies and flexible energy demands
- heat flows characterized by temperature-heat profiles and an intermediary heat transfer network
- the system superstructure, which enables the introduction of any energy service demand or energy production technology

## 4.2.2 Energy symbiosis involving RES in EIPs

The potential for renewable energy use in industry has been widely explored in terms of supporting sustainable development and fostering the low carbon shift in the manufacturing sector. Technical and economic and environmental assessments demonstrate the benefits of possible integration [65]. The main renewable energy sources suitable for industrial application are biomass, solar radiation, ground heat, and wind. The right renewable energy technology or technologies need to be chosen, as well as the size of the plant based on the SMEs' energy demands and profile, and any overproduction handled [66]. The investment efficiency of different RES technologies and the best energy strategy can be evaluated with DEA methods [67,68].

This subsection describes the options and current solutions for RES integration in EIPs. Fig. 4 shows the RES integrated within EIPs according to the reviewed papers: waste/biomass treatment plants, wind turbines, solar panels (thermal collectors and photovoltaic plants), ground-source heat pumps. The carbon capture and storage (CCS) technology, used to prevent carbon emissions in the atmosphere and calculated as negative emissions in the CO<sub>2</sub> emissions assessment, is also shown. Multi-energy systems architectures, including RES, were the most explored and effective option.

Energy symbiosis within EIPs creates exchange networks that can be classified, according to Afshari et al. [63], based on the source of supply. In the first type inter-firm energy waste can be used to supply internal energy demand; in the second type, a set of energy hubs satisfies the energy demand of the involved partners; and in the third type, waste or unused energy (from processes) are shared among companies. According to the authors, this last type of network maximizes the environmental impact reduction.

Thus, heat exchanges between processes can be considered as a first level network designed to optimize energy use; the further step, including exchanges via a central utility system, requires heat exchangers and intermediate fluid [23]. In addition, the excess heat can be supplied through a direct inlet into the district heating network or converted into electric energy by means of a proper conversion technology, avoiding the discharge of heat into the environment [69]. The low temperature heat (below 200°C) is referred to as "waste heat" as it is not directly recoverable in industrial processes. Industrial waste heat can be recovered from flue gas, cooling fluids and exhaust steam. The available and effective technologies to recovery the low-temperature waste heat produced by industrial processes and utilities can be either upgrading technologies, to produce thermal energy for heating or cooling, or converting technologies to convert the waste heat into electric and mechanical power [70]. Industrial heat recovery options are illustrated in Fig. 5.



Fig. 5. Industrial waste heat recovery options (adapted from [70]).

Heat exchangers, heat pipes and condensing boilers, which are upgrading technologies detailed in Fig. 5, are typically used for recovering waste heat that can be reused internally in the industrial process. Heat pumps can be effectively used in the case of district heating. The waste heat recovering for power generation by means of low temperature engines (the Organic Rankine Cycle, Kalina cycle and trilateral cycle) is mainly effective at medium to high temperatures (100°C-200°C).

The recovery of industrial waste heat is suggested by Marchi et al. [71] to improve sustainability of the Brescia (IT) industrial district, through a direct inlet into the district heating network and through electric energy conversion. The presence in the EIP network of a multi-utility operating in the energy sector allows the companies that produce surplus energy (electrical or thermal) to also sell

it to the multi-utility itself for satisfying shared utilities or the local urban area demand. Sludge and waste wood can also be treated to produce fuel.

Organic waste materials are considered the most sustainable biomass supply for energy production, and are a frequently used within industrial parks for recovering waste that cannot be otherwise reused or recycled, thus avoiding landfill ([72], [73]). This includes anaerobic digestion, incineration (direct combustion followed by energy recovery of the heat generated), gasification and pyrolysis. Waste treatment plants are often shared with municipalities so urban waste can also be collected, and represent the most widespread example of synergy between an industrial park and the neighbouring urban area.

The pulp and paper industry is a major industry sector in Finland, and produces waste materials such as bark, wood chips, fibre suspension and milled peat. Sokka et al. [74] evaluate the GHG emissions of the Kymi EIP, where a power plant uses the scraps of the main pulp and paper plant as fuel to generate steam, electricity and heat, which are then delivered to the pulp and paper plant itself, to chemical factories located within the park, and to a regional energy distributor. The authors, through their LCA based analysis, find that emissions would increase by 40-75% if the materials and energy exchange had not been implemented.

Many case studies demonstrate that energy exchange relationships among companies within an EIP lead to a collective GHG emissions reduction. Park et al. [29] reveal that among the 116 implemented during the first phase of the Korean EIP program, 21 projects involved various types of energy recovery such as the use of residual heat from waste incineration or anaerobic digestion and heat recovery from by-products. Ban et al. [75] demonstrate the carbon dioxide direct emissions reduction in commercialized eco-industrial park projects in South Korea, listing the projects aimed at reducing fossil fuel use through the creation of energy networks for energy recovery and waste treatment plants for power generation.

The type of fuel used in shared power plants and heat supply systems significantly affects the level of carbon emissions within the parks [76]. For power plants or energy intensive industries are present, carbon capture and storage technology can help to prevent carbon emissions in the atmosphere. CCS is a recognized technology promoted by the European Union which together with carbon capture and use (CCU) can be used to reduce the carbon footprint of power and industrial sectors. The capturing systems depends on the combustion processes [77] and, after the capturing phase, carbon dioxide can be transported and stored in geological formations such as deep saline aquifers and oil or gas reservoirs. Technologies for reusing captured CO<sub>2</sub> where geographical constraints make storage sites non-viable are explored by Koitsumpa et al. [78]. They present potential uses of CO<sub>2</sub> for fuel and for combined heat and power production in industrial applications.

The introduction of electricity generation plants fuelled by RES within EIPs affects the indirect CO<sub>2</sub> emissions, due to electricity acquired from external suppliers. When also considering the production of thermal energy from renewables, there is also an impact on direct emissions [39]. Guo et al. [79] evaluate the GHG mitigation potential of actions aimed at targeting energy consumption for 213 Chinese industrial parks. The study, through a process-based LCA method that assesses direct and indirect energy-related GHG emissions, shows how the implementation of three measures can bring about significant GHG emission reduction: (i) increasing the share of natural gas and (ii) the efficiency of industrial coal-fired boilers, which have an impact on direct emissions, and (iii) reducing the GHG emission factor of the electricity grid, which has an impact on indirect emissions. The latter can be achieved through low-carbon energy production and upgraded energy infrastructure within industrial parks. Feng et al. [80] compare three low carbon scenarios that can be applied to a Traditional Chinese Medicine industrial park. Assuming that energy conservation measures are implemented, the scenarios include clean energy plants (biomass boilers and photovoltaic (PV) solar energy) with high capacities, which result in a greater carbon emissions reduction, even if the cost effectiveness of such a solution involves the reduction of the solar energy cost.

Despite the major academic efforts to demonstrate the feasibility and effectiveness of the use of renewable energy sources within EIPs, progress is still slow, as reported by Guo et al. [81] in their study involving 106 Chinese EIPs participating in a national demonstration program to facilitate the eco-transformation of industrial parks. They classify the energy infrastructure of the parks by assessing the direct GHG emissions of the power plants, as they represent the main direct GHG emission source. Only 3 EIPs out of 106 share RES power plants: one solar power station with distributed geothermal heat pumps and two solar power stations. In total, the renewable-fuelled power plants (biomass, bioenergy, solar, wind, hydro and geothermal) account for about 1% of the total power generation capacity.

Energy-savings solutions (relighting, insulation) and an increase in electricity and heat produced by means of renewables are suggested by Block et al. [82] to achieve carbon neutrality at the Herdersbrug Industrial Park (Belgium), where the main CO<sub>2</sub> emissions are due to energy consumption and the waste incineration plant. The evaluation shows that about 67% of total CO<sub>2</sub> emissions can be compensated for by the existing and projected renewable energy generated in the park. Renewables plants are already present in the park, in the form of wind turbines and PV panels that cover the roof of two companies (16,000 square meters). The flat or saw-tooth shaped industrial buildings represent a typically unused and exploitable area for installing wind or solar energy generators.

The use of RES is conditional on the local availability of the source and solar resources (including wind) vary with the time of the day, the season and the weather. When the main energy utilization

period is during the daytime, the energy demand matches the supply, maximizing the exploitation of solar energy. The patterns of solar energy utilization for energy supplies in industrial districts located in Yunnan Province in China is investigated by Su et al. [83]. They apply a multi-objective optimization approach to find the optimum solar hybrid energy systems, in which the global warming potential and life-cycle economic cost are minimized, while optimizing the exergy performance of the system. The high cost of electricity for industrial use and the large energy utilization during the daytime leads to the evaluation of solar energy technologies, and particularly low-temperature thermal energy generators, which are more competitive in industrial districts than in residential areas. When the solar energy availability does not match the energy demand, due to energy utilization patterns or weather induced intermittency, energy storage solutions or the use of auxiliary energy technologies are required [84].

An effective method of increasing the RES utilization efficiency at the industrial park level is to combine heat and power generation through the use of combined heat and power (CHP) systems. CHP systems simultaneously generate electricity and useful heat that can be used for heating buildings and supplying hot water. They also allow for the recovery of heat generated by electricity production, resulting in an overall efficiency approaching 90% [85]. CHP plants can be fuelled by various energy sources, including waste and renewable sources such as biogas, biomass and solar and can contribute to diversifying the energy mix of a district. Due to these characteristics, CHP plants support recycling networks and can facilitate both inter-firm cooperation and urban-industrial synergies, and they have long been recognized as meeting the principles of industrial ecology [40].

Solutions such as CHP with the use of ground probes for thermal storage and building integrated photovoltaics (BIPV) can be considered effective options for augmenting energy flexibility in manufacturing environments, and particularly in technical building systems and auxiliary processes, facilitating the use of renewable energy sources and improving the sustainability of industrial processes [12].

The use of various renewable energy sources for generating both electricity and heat (hybrid-RES), and in combination with other generation systems such as tri-generation technologies (combined cooling, heat and electricity), energy storage systems and energy distribution networks, enable the configuration of a smart multi-energy system (SMES or smart-MES) as represented in Fig. 6. Smart-MES can be modelled to support decision-makers in identifying and choosing the better generation options including RES [47]. The concept of smart-MES (sometimes called Smart Multi-Energy Grids) extends the concept of the smart-grid [86], typically defined within the limitations of the electricity sector, by integrating multi energy carriers. Applying the smart-grid architecture enables the discontinuity of renewable sources such as solar or wind to be managed. The main characteristics of smart grids are the use of ICT and smart technologies for controlling and managing the grids and the integration of distributed energy resources (DER). DER are considered to be the main pathway towards an effective integration of discontinuous sources into the energy system, and involve small-scale units of local generation connected to the grid at the distribution level. The DER configuration typically includes storage solutions and demand-side resources (load management systems, energy efficiency options), and allows a two-way flow of power between the decentralized grid and the main distribution grid [87]. Depending on the energy demand end-uses the optimal combination of DER technologies can be evaluated using DEA models, to compare and calculate the factor efficiencies of each energy technology for each end use [88].



**Fig. 6**. The smart multi-energy system concept, enabling the smart control and management of the grids and optimizing the use of distributed energy resources.

EIPs internal utility networks can be viewed as small-scale grids and, as they are generally interfaced with the main utility network through a single point of common coupling, they can be considered as microgrids. Industrial smart microgrids, consisting of inter-connected loads and DER including RES, can be developed and operated in a controlled and coordinated way, optimizing the control of the individual units and the grid itself. A smart MES configuration promotes a coordinated energy strategy within EIPs. According to Deckmyn et al. [89], within the energy management system, the microgrid control architecture can be centralized or decentralized. A centralized controller optimizes the exchanged power between the microgrid and the utility grid by maximizing the local production, gathering data from every DER within the microgrid; it is highly efficient, but due to the complexity of the system a single point of failure may arise. A decentralized management system uses local controllers that work autonomously, controlling specific DER units, but in a coordinated way with other local controllers to improve the overall performance of the microgrid. They maximize their production to satisfy the demand and provide the maximum

possible export to the grid. Within industrial parks the loads have different owners, as may the generation and storage units, so decentralized control can be the preferred solution.

The energy management system of a micro-grid consists of a supervisor [90] who optimizes the power flow within the system. The first stage involves forecasting the inputs required; then the system solves the optimization problem by establishing the state of all the equipment; lastly it sends control signals to the power generation devices. Many micro-grid energy management systems have been studied in the literature [91], employing various decision making strategies and solutions.

Reynolds et al. [92] suggested using multiple artificial neural networks (ANNs) to perform both supply and demand optimization at the district level in order to manage demand-side flexibility in a smart multi-energy context. An ANN is used for weather forecasts and the prediction of the energy consumption [93] and power output of generation sources [94]. ANN methods are widely used to manage RES intermittency within the distribution system [95], and are applied in forecasting, design optimisation, fault detection and optimal control [96]. They estimate/forecast solar-related variables (such as solar radiation and clearness index), atmospheric-related variables (such as wind speed), hydrologic-related variables and geologic-related variables. The combination of these variables enables the optimal management of RES [97] and the balance of energy exchanges with the main distribution grid [98].

A suitable tool for the integrated management of a smart MES can be modelled as an energy hub (EH). Mohammadi et al. [99] defined this as "the place where the production, conversion, storage and consumption of different energy carriers takes place". An EH is in essence an interface between primary energy sources and end-users, incorporating energy conversion and storage processes. A typical energy hub uses input energy carriers (electricity and natural gas), energy converters (transformers, gas turbines, gas boilers, electrical chillers and absorption chillers, and RES converters), energy storage devices and provides electricity in addition to heating and cooling energy services as the outputs. A schematic diagram of the energy hub concept is depicted in Fig. 7.



Fig. 7. The energy hub (adapted from [99,100])

As Fig. 7 shows, within the EH different energy sources are converted using suitable technologies. The efficient use of multi-generation systems allows the use of the energy resources to be optimized, increasing the efficiency and reducing emissions and costs [101]. A similar concept to EH is outlined by Xing et al. [102] in the modelling of a distributed energy system. They applied a multi-objective method to the design of a DER system within an EIP, considering the natural gas allocation scheme between industrial parks to manage gas shortages.

Another aggregation concept for DERs is the Virtual Power Plant (VPP), a cloud-based control centre that uses communication technologies to gather data from distributed power plants, controlling and managing generated power and energy flows. It emulates the functions of a traditional power plant, enabling small distributed energy resources to participate in the energy market. Although this approach had been developed mainly for addressing distributed electrical resources managing, it is also well suited to multi-energy applications [103].

# 4.2.3 Energy organizational strategies within EIPs

As discussed in the previous subsection, a multi-dimensional mix of technological options is available to improve the energy efficiency and autonomy of an EIP. Energy efficiency measures, both at the industrial operational level (more efficient processes and machineries, introduction of heat exchangers and fuel switching to renewables) and the buildings level, and energy conversion systems using available renewable sources are effective choices for reducing the amount of imported energy into the park. In addition, if a joint planning strategy and management is developed, the interventions at park level can result in greater GHG emission reductions than would be possible through individual reduction interventions [104].

The key factors for facilitating the distribution of renewable energy technologies are the information available to potential adopters, the interaction of involved actors and the existence of a critical mass (firms' agglomeration) to reduce costs. In addition, the concentration of many firms in the same location enables them to take advantage of common services and common service providers [14]. Thus, the planning and managing of EIPs should include a collective energy strategy stimulating the use of renewable energy, through the purchase of RES generated electricity, individual or collective self-production of green electricity, energy cooperation among industries through infrastructures sharing [30]. Timmerman et al. [64] present an EIP energy system scheme where individual companies can be either connected to energy conversion units or to an internal energy network with storage facilities, supplying a number of companies. The local network can be connected to both the regional distribution grid and the district heating (DH) network.

Feng et al. [80] introduce the concept of the zero carbon industrial park (ZEIP), where the inventory of energy demand and supply, carbon emissions, and negative emissions, are considered. In addition to direct and indirect carbon emissions, a carbon offset term is introduced, to account for clean energy supply, energy conservation, and negative emissions (e.g. carbon capture and storage), and, once the park energy consumption and emissions are known, the energy strategy can be designed to maximize the carbon offset. Within the proposed framework for the implementation of the ZEIP scheme, the management and technical levels are separated. The management level includes the selection of the participant enterprises, the energy management, and the monitoring of energy consumption and carbon emissions. The technical level includes the technical measures to improve energy efficiency, balance energy demand and supply and balance carbon emissions and offset. The technical measures aimed at maximising the carbon offset are illustrated in Fig. 8.



## Fig. 8. Carbon offset maximization technical measures (adapted from [80])

The selection of the technology to be included in the carbon offset scheme is based on a comprehensive evaluation of profitability, the carbon reduction potential and the social impacts. The introduction of RES to improve the carbon offset can be managed by a central energy server, which is responsible for technical guidance, system management, in addition to equipment and grid maintenance. Energy conservation includes a wide range of measures such as improved process efficiency, energy recovery and energy saving solutions. Negative emissions are calculated for plantations.

Hentschel et al. [105] discuss the introduction of renewable energy cooperatives, as a form of clustering involving firms and other local stakeholders to foster the deployment of renewable energies within industrial clusters. A renewable energy cooperative composed of companies can be seen as a strategic alliance, in which they can manage market uncertainty linked to RES investments, improve energy efficiency, reduce their dependence on external energy suppliers, get

together the required skills and resources, and exploit the available solar energy. The study identifies the specific characteristics of a cooperative required for facilitating the energy transition of the Port of Rotterdam, one of Europe's major industrial clusters, but it can also serve as support for other clusters' stakeholders. Organizational issues, clear targets and milestones, trust and close communication among partners are the main attributes required for a successful project. There needs to be a trade-off among the potentially conflicting objectives of the stakeholders involved in creating energy symbiosis, for example reducing investment costs vs reduced environmental impact [106].

The benefits of setting an energy strategy at EIP level are summarized in [30] and [38]. Energy clustering can provide many benefits for participating companies: it enables the investment costs for plants and infrastructures installation to be reduced, along with the operational expenditure (fuel, maintenance). It also provides favourable prices for collectively purchased utilities and a more even load curve by bundling the energy demands of the different firms, reduces their dependence on energy market prices due to the sharing of the energy produced, and results in reduced taxes and improved brand images, due to the lower carbon footprint.

## 4.3 Urban-industrial energy symbiosis

In addition to the industrial symbiosis concept, the concept of urban-industrial symbiosis has emerged. In their analysis of the development of IS and the related practice in Japan, Van Berkel et al. [41] identify the emerging of the urban-industrial symbiosis concept in relation to the Japanese Eco-Town Program. This Program (which started in 1997 and ended in 2006) extended the focus of industrial sustainability from industry eco-efficiency initiatives to industrial symbiosis and urban-industrial synergies.

Some types of industrial sites are equipped with waste/sludge treatment facilities that can also use urban waste as fuel, improving plant saturation and reducing waste landfill and thus bringing advantages both to industry and neighbouring urban areas. This kind of synergy is the basic UIS approach. Urban areas deliver urban solid waste for incineration or recycling, urban organic waste for energy production (e.g., by anaerobic digestion, pyrolysis, and gasification) [107], waste water for treatments, sewage sludge for energy production [74], and receive back electrical or thermal energy from industrial clusters. However, urban areas also play an important role in energy consumption and carbon emissions and EIPs can provide clean energy to local communities, thus improving the social role of industry [19].

Different approaches to urban-industrial symbiosis emerge from the literature. Industrial parks can be located near already established cities, or as in China, rapid industrialization starting from the 1980s lead to the establishment of large-scale industrial districts followed by the growth of related urban districts, which now face strong environmental degradation. A Chinese national EIP program

was launched in 2001, supporting the eco-transformation of industrial parks towards comprehensive eco-cities, mainly through the implementation of energy efficiency and pollution reduction measures. During the 2010s, low carbon and sustainability issues encouraged further interest in urban-industrial energy synergies.

Dong et al. [19] analyse the advantageous relationships between industrial parks and urban communities that have achieved a successful low-carbon strategy. By focusing on ongoing projects in China, they note that material symbiosis (mainly urban waste recycling systems) is more common than energy symbiosis, mainly due to the physical characteristics of energy being more unstable to transport and requiring costly transportation facilities. However, the further opportunity to reduce GHGs emissions in industrial areas in China makes urban-industrial energy symbiosis a solution that is also assessed and promoted through policy support, as technology innovations can enable more energy efficient transportation, and information platforms enable the collection and exchange of information regarding material/energy/waste flows. The case study they investigate, regarding the urban-industrial symbiosis at the cities of Jinan and Liuzhou, demonstrate the CO<sub>2</sub> emissions reduction potential of this type of project.

The urban-industrial symbiosis of the Suzhou Industrial Park and Suzhou City energy efficiency solutions, in combination with the funded integration of clean and renewable energy solutions (such as CHP, water/ground source heat pumps, solar water heaters), led to clean energy accounting for 78.6% of the total usage in 2012 [108].

The Eco-town program developed in Japan aimed at promoting advanced urban planning in accordance with the zero-emission concept: in particular, it fostered industry synergies focusing on the reduction of industrial waste and material recycling and supported the development of plastic and electrical appliance recycling facilities in many eco-towns. Only two Eco-towns [76], Kawasaki and Kitakyushu, feature energy exchange by using waste heat from the industrial district.

Kim et al. [109] focus on energy symbiosis networks, analysing a urban-industrial symbiosis scenario linking the Onsan and Ulsan-Mipo national industrial parks to Ulsan city (South Korea), to evaluate the environmental and economic benefits of re-using industrial waste heat at both industrial and urban levels. Two recovery systems are proposed: a pipeline for waste steam exchanges between industries and a central heating system collecting the industrial waste heat for heating and cooling purposes of the residential and commercial area. The potential of sending municipal waste to the incinerator plant to generate heat for industrial use and to a digester to produce methane for fuelling electricity generators is then considered to complete the UIS framework. Environmental and economic benefits can be demonstrated through this energy symbiosis by sharing the costs of the initial investment at EIP and public levels.

Focusing on the Fukushima region after the 2011 earthquake, Togawa et al. [69] discuss the need for the effective recovery of underused thermal energy, like waste heat, to provide sustainable energy sources that reduce the environmental impact on local urban areas. They consider the concept of energy symbiosis essential at the regional energy planning level, based on the idea of industrial symbiosis, for implementing energy exchanges between factories and between factories and residential/commercial areas, in a district heating (DH) model. In the consideration of heat transfer systems, the proximity of energy consumers to suppliers is a key design factor.

District heating is a flexible technology enabling the use of many fuels or sources of heat; it comprises a network of pipes connecting the buildings to be heated and/or cooled, which can be served by centralized plants or distributed units, such as CHP systems, heat pumps, geothermal and solar thermal units, heat from waste-to-energy plants or from industry thermal energy surplus. The use of thermal storage solutions allows solar sources to be closely integrated and can be useful for combining different heat sources in the network [110].

There is a long history of research and applications into district heating networks, especially in Baltic and Nordic countries where DH supplies more than 60% of the total, while more than 21% of Austrian households are heated by DH systems. Karner et al. [18] provide an overview of the possible synergies between industries and nearby towns, considering that excess industrial energy can cover a part of the energy demand of the urban area. They include in the industrial energy available for urban recovery typical industrial heat and waste streams (both high temperature and under-90°C waste heat, warm wastewater, and waste), and introduce the exploitation of industrial empty roof space, in which photovoltaic plants for renewable power generation can be installed, as summarized in Table 4.

Energy carrier	Usage	Technology	Temperature range
Waste beat	Heat, electricity,	Heat pump (heating, cooling)	< 90 °C
Waste fleat	cooling	Heat exchanger	> 90°C
		Rankine cycle	> 90°C
Wastewater	Heat	Heat pump	35-90°C
		Heat exchanger	> 90°C
Waste	Heat, electricity	Heating plant	> 90°C
		CHP, Rankine cycle	> 90°C
Roof areas	Electricity	Photovoltaic	

		Heat	Solar thermal	< 200°C
--	--	------	---------------	---------

**Table 4**. Possible energy carriers and usage in a urban-industrial context [18].

The model developed for industrial energy usage in urban contexts, which was applied to four urban regions in Austria, shows that up to 35% of the total urban energy demand, and 6 to 46% of the electricity demand, could be supplied by industrial energy thus lowering CO<sub>2</sub> emissions. By analysing the economic viability of integrating industrial energy into urban energy systems in these urban areas [111], different business models for financing and operating plants providing industrial energy can be distinguished, as detailed in Table 5.

Industrial	Financed by	Operated by	Annual	Notes
energy			revenues	
Electricity: on- roof PV plant	The company itself	The company itself	Feed-in of total electricity production (feed- in tariffs) Feed-in of excess electricity production (feed- in tariffs; relevant for small industrial units and high electricity prices)	Rather unlikely as the capital commitment and the amortization time are too long
	External company	External company	Green certificates	The company roof is rented
	Citizens participation	External company	Feed in tariffs	The company roof is rented
	Partly by crowdfunding (equity) and the remaining share through banks	External company	Feed in tariffs	Higher interest rate of the publicly financed equity
Heat: delivered	The company	The company		Rather unlikely since the capital

through DH-grid	itself	itself	commitment and
			the amortization
			time are too long
	A service	A service	
	company	company	

|--|

It should be noted that both the installation and operating phases of power generation plants are not part of the main businesses of companies, so the preferred option is to rely upon external service providers. Based on the different scenarios and as profits for industrial companies and energy service providers must be guaranteed, it was found that the usage of industrial heat is economically feasible in any scenario, while the industrial electricity generated from roof-top PV is only profitable if feed-in tariffs are received, and is significantly influenced by the cost of PV modules. A threshold value determines the profitability of building a waste incineration plant.

A smart energy system approach is suggested by Lund et al. [86], in which smart electricity, thermal energy and gas grids are combined with storage technologies and coordinated, providing synergies among different sectors (industrial, commercial and residential) to achieve optimal sustainable solutions for each individual sector and for the whole system. The benefits for the local communities can be evaluated in relation to environmental restoration and improved well-being, while eco-efficiency and eco-innovation are the main advantages for the involved industries [41]. A regional multi-energy "prosumers" (both energy producers and consumers) scheme based on interconnected energy hubs combining distributed energy supply, RES and CHP, are modelled in [100] to serve residential, commercial and industrial districts.

Afshari et al. [112] propose a mathematical model to extend industrial energy symbiosis to residential areas. In the analysed energy networks scheme, industries share their waste and unused energy with partners, i.e., electricity suppliers, and both industrial and residential energy users. The model determines the potential synergies by minimizing the total cost and environmental impact of an energy exchanger network. It also evaluates the effects of uncertainties on the symbioses over the long term, such as variations in energy demand and supply and in price and considers the concerns of residential and industrial users in the decision-making process. The example shows that improved cost savings and pollution reduction of optimal networks can be obtained by industrial-residential symbiosis compared to industrial symbiosis. The involvement of urban residents in the feasibility study and in the implementation and operation phases of the urban-industrial symbiosis is suggested. A strong interaction between the urban residents and the business park is considered a characteristic of a mixed-use ecopark [113], highlighting the need for a collective awareness and effort in implementing sustainable strategies.

As emerging smart technologies support the transition to low-carbon lifestyles and business patterns, more integrated smart infrastructures and energy management platforms between smart cities and industrial parks will promote the use of renewable energy and urban industrial symbiosis [114].

## 5. Conceptual framework and a research agenda

The aim of this article is to provide a conceptual synthesis of collective energy strategies and the integration of EIPs and RES, and to design a future research agenda. The importance of developing collective energy strategies has been demonstrated by many authors, who suggest options highlighting the role of EIP and the use of RES at both industrial and urban levels. The review includes a discussion and descriptions of the possibilities, options, and existing solutions for collective energy strategies. Fig. 9 shows the conceptualization framework constructed for the articles retrieved from the literature.



Fig. 9. Conceptual framework and articles focusing on each key element.

Fig. 9 presents articles that support each key element. The scheme emphasizes four main pathways to implement collective energy strategies within an EIP: the collective purchase of green

energy, traditional energy exchanges and recovery, the collective production management of energy, and shared building services and utilities.

Collective purchase of green energy covers joint contracting of electricity produced from renewable sources. Traditional energy exchanges and recovery include the use of surplus energy, such as heat exchanges between processes or exchanges via a central utility system, and also involve heat exchangers and intermediate fluids. This category includes waste/sludge treatment facilities for energy production. The term traditional refers to the classical approach to IS. Collective energy production covers collective large-scale plants for the production of clean energy, generating both electrical and thermal energy, and distributed multi-generation units. Both possibilities rely on the integration of low carbon energy (including CHP units and CCS solutions) and the extensive use of RES. Sharing of building services and utilities covers RES integration (e.g., BPIV) for solutions following a nearly Zero-Energy Buildings approach. A wider energy strategy can include energy exchanges between industrial and urban areas, thus developing urban-industrial energy symbiosis.

The conceptual framework developed in this review supports the design of a research agenda for a collective energy strategy to promote RES uptake at the industrial and urban levels. Some of the key elements identified do not consider the use of RES (traditional energy exchanges

and recovery) or would need a specific in-depth analysis to suggest further research opportunities (such as the sharing of building services and utilities).

Future research could include:

(a) simulations for collective energy strategies: while the reduction in GHG emissions as a result of energy symbiosis has been widely demonstrated, we found a few analyses that demonstrate the applicability of such strategies, which can provide technical evaluation tools and raise the awareness of the actors involved in EIP development;

(b) surveys for collective energy production: our review focused on the energy strategies implemented in EIPs. A survey on collective energy production strategies, involving low carbon technologies, would facilitate generalization of the study results for the initial design of EIP development schemes;

(c) case studies and subsequent action research for distributed energy systems in EIPs, to obtain results from the application of qualitative and quantitative modelling and evaluation of different technological and organizational strategies. Although in the reviewed papers, the energy exchanges (mainly heat) within EIPs have been extensively modelled and studied, few papers apply the distributed energy systems scheme to EIPs and only one case study have been found [102]. EIPs represent a suitable platform for modelling and testing multi energy systems.
(d) there is a wide research gap in terms of integrating RES in urban-industrial energy symbiosis: the advantages of energy systems is still under-investigated. RES integration through distributed

architectures, often separately analysed in terms of EIPs and at the community level (urban areas) [115], also needs further investigations. In this context, energy management plays a critical role in overall efficiency improvements as well as in CO<sub>2</sub> emission reduction [116,117] Therefore, energy efficiency management in energy-intensive industries [118,119] as well as machine-learning techniques for energy management [120,121] can also be studied in in-depth case studies, associated with RES networks.

## 6. Conclusions

The industry sector is urged to reduce carbon emissions by materials and energy efficiency measures. Replacing fossil fuels with renewable energy sources would help to reduce carbon emissions at the industrial level. However, individual firms prioritise value-adding processes over energy related projects that require specific expertise, which is often not available internally. Cooperation in industrial parks can help overcome the lack of technical knowledge regarding low carbon and renewable technologies at a reasonable cost, by collectively consulting a service provider. Eco-industrial parks, which were set up to reduce the environmental footprint of the firms involved, represent a suitable cooperative model for fostering the integration of RES in the industrial system.

The paper presents the results of a literature review that has explored the synergetic possibilities of integrating EIP and RES, enabling the uptake of renewable energy sources at an industrial level and potentially creating urban-industrial energy symbiosis.

Three main themes emerged from the reviewed papers: i) the sustainability of energy projects within EIPs; ii) energy symbiosis schemes within EIPs promoting RES integration; and iii) urbanindustrial energy symbiosis.

We believe that this review contributes to the literature by providing an original perspective on the sustainability potential of EIPs focused on energy symbiosis. We have organized knowledge on energy sustainability in industrial parks within a conceptual framework. This framework demonstrates viable solutions and the effectiveness of RES integration within EIPs for reducing the carbon footprint of industry, and potentially creating urban-industrial energy symbiosis.

This study involved a review and analysis of papers on energy symbiosis strategies within ecoindustrial parks, both considering the influence of EIP models on energy issues and the use of renewable energy sources. In addition, it reviews and analyses the urban-industrial energy symbiosis approaches originating from EIP energy management solutions which include RES.

The possibilities, options, and existing solutions for collective energy strategies within EIPs were analysed, and a conceptualization framework was presented, which shows four main pathways for

creating collective energy strategies. In addition to the classical IS approach consisting of inter-firm energy exchanges, the collective purchase of green energy, the collective production management of energy, and shared building services and utilities are viable solutions for improving RES uptake at the industrial level.

The study has resulted in a conceptualization framework demonstrating four main pathways to implement energy synergies within EIPs and viable solutions to improve the uptake of renewable energy sources at the industrial level. The conceptual framework supports the design of a future research agenda by revealing the most promising research opportunities as key elements. The main gap identified in the research is that the advantages of energy symbiosis networks between industrial and urban areas that also integrate renewable energy systems have not been fully investigated.

We believe that this paper provides researchers, technicians, industry managers and decision makers involved in energy managing and sustainability issues, with an outlook of the implemented and possible options to promote GHG mitigation in the industry sector with the aim of both informing and providing suggestions for further studies.

The key points of this review are:

o An analysis of energy symbiosis schemes within eco-industrial parks, considering the influence of EIP models on energy concerns and the use of renewable energy sources as viable options to reduce industry GHG emissions

o An analysis urban-industrial energy symbiosis approaches originating from EIP energy management solutions that include renewable energy sources

o A conceptual synthesis, showing four main pathways for creating suitable collective energy strategies aimed at improving RES uptake at the industrial level

o Several future research opportunities, revealing that the advantages of energy symbiosis networks between industrial and nearby urban areas that also integrate renewable energy systems are still under-investigated.

The main limitation of the research is due to the spreading of research field related to the topic under study. The study thus represents a first synthesis of the knowledge in the fields of energy symbiosis, eco-industrial parks and renewable energy sources. A more in-depth analysis of the literature should be performed on more specifically energy-related themes which can support the implementation of energy symbiosis schemes involving RES. Future studies could review the methods for energy analysis and optimization, analysing the suitability for energy strategies within EIPs and in urban-industrial energy symbiosis projects. Applying these methodologies to case studies would also be valuable.

As previously mentioned, the main challenges identified by this study require future experimental research, in terms of applying the collective energy strategies to specific case studies.

## References

- Johnson NE, Ianiuk O, Cazap D, Liu L, Starobin D, Dobler G, et al. Patterns of waste generation: A gradient boosting model for short-term waste prediction in New York City. Waste Manag 2017;62:3–11. doi:10.1016/j.wasman.2017.01.037.
- [2] Gereffi G, Lee J. Economic and Social Upgrading in Global Value Chains and Industrial Clusters: Why Governance Matters. J Bus Ethics 2016;133:25–38. doi:10.1007/s10551-014-2373-7.
- [3] Zhu Q, Qu Y, Geng Y, Fujita T. A Comparison of Regulatory Awareness and Green Supply Chain Management Practices Among Chinese and Japanese Manufacturers. Bus Strateg Environ 2017;26:18–30. doi:10.1002/bse.1888.
- [4] Ma L, Chong C, Zhang X, Liu P, Li W, Li Z, et al. LMDI decomposition of energy-related CO2 emissions based on energy and CO2 allocation Sankey diagrams: The method and an application to China. Sustain 2018;10. doi:10.3390/su10020344.
- [5] Nastasi B, Lo Basso G. Hydrogen to link heat and electricity in the transition towards future Smart Energy Systems. Energy 2016;110:5–22. doi:10.1016/j.energy.2016.03.097.
- [6] United Nations Industrial Development Organization. Industrial Development Report 2018.
   Demand for Manufacturing: Driving Inclusive and Sustainable Industrial Development.
   Vienna: 2017.
- [7] International Energy Agency. World Energy Outlook 2017 Summary. 2017.
- [8] IEA/OECD, Cédric Philibert. Renewable Energy for Industry From green energy to green materials and fuels. 2017. doi:10.1111/j.1365-2990.2010.01130.x.
- [9] Butturi, M.A., Lolli, F., Balugani, E., Gamberini, R., Rimini B. Distributed renewable energy generation: A critical review based on the three pillars of sustainability. Proc Summer Sch Fr Turco Vol 2018-September, 2018 2018;2018:179–85.
- [10] Paudyal S, Canizares CA, Bhattacharya K. Optimal Operation of Industrial Energy Hubs in Smart Grids. IEEE Trans Smart Grid 2015;6:684–94. doi:10.1109/TSG.2014.2373271.
- [11] Ozorhon B, Batmaz A, Caglayan S. Generating a framework to facilitate decision making in renewable energy investments. Renew Sustain Energy Rev 2018;95:217–26. doi:10.1016/j.rser.2018.07.035.
- [12] Weeber M, Lehmann C, Böhner J, Steinhilper R. Augmenting Energy Flexibility in the Factory Environment. Procedia CIRP 2017;61:434–9. doi:10.1016/j.procir.2016.12.004.
- [13] International Energy Agency. Accelerating Energy Efficiency in Small and Medium-sized Enterprises. Powering SMEs to catalyse economic growth. 2015.
- [14] Horbach J, Rammer C. Energy transition in Germany and regional spill-overs: The diffusion of renewable energy in firms. Energy Policy 2018;121:404–14. doi:10.1016/j.enpol.2018.06.042.
- [15] Bellantuono N, Carbonara N, Pontrandolfo P. The organization of eco-industrial parks and

their sustainable practices. J Clean Prod 2017;161:362–75. doi:10.1016/j.jclepro.2017.05.082.

- [16] Sellitto MA, Luchese J. Systemic Cooperative Actions among Competitors: the Case of a Furniture Cluster in Brazil. J Ind Compet Trade 2018;18:513–28. doi:10.1007/s10842-018-0272-9.
- [17] Côté RP, Cohen-Rosenthal E. Designing eco-industrial parks: a synthesis of some experiences. J Clean Prod 1998;6:181–8. doi:10.1016/S0959-6526(98)00029-8.
- [18] Karner K, Theissing M, Kienberger T. Energy efficiency for industries through synergies with urban areas. J Clean Prod 2016;119:167–77. doi:10.1016/j.jclepro.2016.02.010.
- [19] Dong L, Gu F, Fujita T, Hayashi Y, Gao J. Uncovering opportunity of low-carbon city promotion with industrial system innovation: Case study on industrial symbiosis projects in China. Energy Policy 2014;65:388–97. doi:10.1016/j.enpol.2013.10.019.
- [20] Fang K, Dong L, Ren J, Zhang Q, Han L, Fu H. Carbon footprints of urban transition: Tracking circular economy promotions in Guiyang, China. Ecol Modell 2017;365:30–44. doi:10.1016/j.ecolmodel.2017.09.024.
- [21] Jiao W, Boons F. Toward a research agenda for policy intervention and facilitation to enhance industrial symbiosis based on a comprehensive literature review. J Clean Prod 2014;67:14–25. doi:10.1016/j.jclepro.2013.12.050.
- [22] Boix M, Montastruc L, Azzaro-Pantel C, Domenech S. Optimization methods applied to the design of eco-industrial parks: A literature review. J Clean Prod 2015;87:303–17. doi:10.1016/j.jclepro.2014.09.032.
- [23] Kastner CA, Lau R, Kraft M. Quantitative tools for cultivating symbiosis in industrial parks; a literature review. Appl Energy 2015;155:599–612. doi:10.1016/J.APENERGY.2015.05.037.
- [24] Jenniches S. Assessing the regional economic impacts of renewable energy sources A literature review. Renew Sustain Energy Rev 2018;93:35–51. doi:10.1016/j.rser.2018.05.008.
- [25] Dong L, Liang H, Zhang L, Liu Z, Gao Z, Hu M. Highlighting regional eco-industrial development: Life cycle benefits of an urban industrial symbiosis and implications in China. Ecol Modell 2017;361:164–76. doi:10.1016/j.ecolmodel.2017.07.032.
- [26] Wang L, Wei Y-M, Brown MA. Global transition to low-carbon electricity: A bibliometric analysis. Appl Energy 2017;205:57–68. doi:10.1016/j.apenergy.2017.07.107.
- [27] Chertow M, Park JY. Scholarship and practice in industrial symbiosis: 1989-2014. In: Clift R, Drukman A, editors. Tak. Stock Ind. Ecol., Springer; 2015, p. 87–116. doi:10.1007/978-3-319-20571-7.
- [28] Saavedra YMB, Iritani DR, Pavan ALR, Ometto AR. Theoretical contribution of industrial ecology to circular economy. J Clean Prod 2018;170:1514–22. doi:10.1016/J.JCLEPRO.2017.09.260.

- [29] Park JM, Park JY, Park HS. A review of the National Eco-Industrial Park Development Program in Korea: Progress and achievements in the first phase, 2005-2010. J Clean Prod 2016;114:33–44. doi:10.1016/j.jclepro.2015.08.115.
- [30] Maes T, Van Eetvelde G, De Ras E, Block C, Pisman A, Verhofstede B, et al. Energy management on industrial parks in Flanders. Renew Sustain Energy Rev 2011;15:1988– 2005. doi:10.1016/j.rser.2010.11.053.
- [31] Susur E, Hidalgo A, Chiaroni D. A strategic niche management perspective on transitions to eco-industrial park development: A systematic review of case studies. Resour Conserv Recycl 2019;140:338–59. doi:10.1016/j.resconrec.2018.06.002.
- [32] Merli R, Preziosi M, Acampora A. How do scholars approach the circular economy? A systematic literature review. J Clean Prod 2018;178:703–22.
   doi:10.1016/j.jclepro.2017.12.112.
- [33] Masi D, Day S, Godsell J. Supply chain configurations in the circular economy: A systematic literature review. Sustain 2017;9. doi:10.3390/su9091602.
- [34] Winans K, Kendall A, Deng H. The history and current applications of the circular economy concept. Renew Sustain Energy Rev 2017;68:825–33. doi:10.1016/j.rser.2016.09.123.
- [35] Valenzuela-Venegas G, Salgado JC, Díaz-Alvarado FA. Sustainability indicators for the assessment of eco-industrial parks: classification and criteria for selection. J Clean Prod 2016;133:99–116. doi:10.1016/j.jclepro.2016.05.113.
- [36] Abas N, Kalair A, Khan N. Review of fossil fuels and future energy technologies. Futures 2015;69:31–49. doi:10.1016/j.futures.2015.03.003.
- [37] Schleicher-Tappeser R. How renewables will change electricity markets in the next five years. Energy Policy 2012;48:64–75. doi:10.1016/j.enpol.2012.04.042.
- [38] Fichtner W, Frank M, Rentz O. Inter-firm energy supply concepts: an option for cleaner energy production. J Clean Prod 2004;12:891–9. doi:10.1016/j.jclepro.2004.02.036.
- [39] Aro T. Preconditions and tools for cross-sectoral regional industrial GHG and energy efficiency policy-A Finnish standpoint. Energy Policy 2009;37:2722–33. doi:10.1016/j.enpol.2009.03.005.
- [40] Korhonen J. Co-production of heat and power: an anchor tenant of a regional industrial ecosystem. J Clean P 2001;9:509–17. doi:10.1016/S0959-6526(01)00009-9.
- [41] Van Berkel R, Fujita T, Hashimoto S, Geng Y. Industrial and urban symbiosis in Japan: Analysis of the Eco-Town program 1997-2006. J Environ Manage 2009;90:1544–56. doi:10.1016/j.jenvman.2008.11.010.
- [42] Chertow M, Ehrenfeld J. Organizing Self-Organizing Systems: Toward a Theory of Industrial Symbiosis. J Ind Ecol 2012;16:13–27. doi:10.1111/j.1530-9290.2011.00450.x.
- [43] Chertow MR. I NDUSTRIAL S YMBIOSIS : Literature and Taxonomy. Annu Rev Energy Environ 2000;25:313–37. doi:doi:10.1146/annurev.energy.25.1.313.

- [44] Frosch RA, Gallopoulos NE. Strategies for Manufacturing. Sci Am 1989;261:144–52. doi:10.1038/scientificamerican0989-144.
- [45] Domenech T, Bleischwitz R, Doranova A, Panayotopoulos D, Roman L. Mapping Industrial Symbiosis Development in Europe\_ typologies of networks, characteristics, performance and contribution to the Circular Economy. Resour Conserv Recycl 2019;141:76–98. doi:10.1016/j.resconrec.2018.09.016.
- [46] Lambert AJD, Boons FA. Eco-industrial parks: Stimulating sustainable development in mixed industrial parks. Technovation 2002;22:471–84. doi:10.1016/S0166-4972(01)00040-2.
- [47] Simeoni P, Nardin G, Ciotti G. Planning and design of sustainable smart multi energy systems. The case of a food industrial district in Italy. Energy 2018;163:443–56. doi:10.1016/j.energy.2018.08.125.
- [48] Kim HW, Ohnishi S, Fujii M, Fujita T, Park HS. Evaluation and Allocation of Greenhouse Gas Reductions in Industrial Symbiosis. J Ind Ecol 2018;22:275–87. doi:10.1111/jiec.12539.
- [49] Martin M, Harris S. Prospecting the sustainability implications of an emerging industrial symbiosis network. Resour Conserv Recycl 2018;138:246–56.
   doi:10.1016/j.resconrec.2018.07.026.
- [50] ISO 14045-2012 Environmental management Eco- efficiency assessment of product systems Principles, requirements and guidelines 2012.
- [51] Fan Y, Bai B, Qiao Q, Kang P, Zhang Y, Guo J. Study on eco-efficiency of industrial parks in China based on data envelopment analysis. J Environ Manage 2017;192:107–15. doi:10.1016/j.jenvman.2017.01.048.
- [52] Pai J Te, Hu D, Liao WW. Research on eco-efficiency of industrial parks in Taiwan. Energy Procedia 2018;152:691–7. doi:10.1016/j.egypro.2018.09.232.
- [53] Wang Q, Sun Y, Yuan X, Cao D, Zuo J, Gao Z. Addressing the efficiency of the core ecological industrial chain: A DEA approach. J Clean Prod 2017;156:235–43. doi:10.1016/j.jclepro.2017.03.185.
- [54] Liu Z, Adams M, Cote RP, Geng Y, Ren J, Chen Q, et al. Co-benefits accounting for the implementation of eco-industrial development strategies in the scale of industrial park based on emergy analysis. Renew Sustain Energy Rev 2018;81:1522–9. doi:10.1016/j.rser.2017.05.226.
- [55] Ehrenfeld J, Gertler N. Industrial Ecology in Practice. The evolution of interdependence at Kalundborg. J Ind Ecol 1997;1:67–79.
- [56] Heeres RR, Vermeulen WJ V, De Walle FB. Eco-industrial park initiatives in the USA and the Netherlands: First lessons. J Clean Prod 2004;12:985–95. doi:10.1016/j.jclepro.2004.02.014.
- [57] Huang B, Yong G, Zhao J, Domenech T, Liu Z, Chiu SF, et al. Review of the development of

China's Eco-industrial Park standard system. Resour Conserv Recycl 2019;140:137–44. doi:10.1016/j.resconrec.2018.09.013.

- [58] Pilouk S, Koottatep T. Environmental performance indicators as the key for eco-industrial parks in Thailand. J Clean Prod 2017;156:614–23. doi:10.1016/j.jclepro.2017.04.076.
- [59] Wang Q, Lu S, Yuan X, Zuo J, Zhang J, Hong J. The index system for project selection in ecological industrial park: A China study. Ecol Indic 2017;77:267–75. doi:10.1016/j.ecolind.2017.01.032.
- [60] Massard G, Leuenberger H, Dong TD. Standards requirements and a roadmap for developing eco-industrial parks in Vietnam. J Clean Prod 2018. doi:10.1016/j.jclepro.2018.03.137.
- [61] Fabrizio E, Branciforti V, Costantino A, Filippi M, Barbero S, Tecco G, et al. Monitoring and managing of a micro-smart grid for renewable sources exploitation in an agro-industrial site. Sustain Cities Soc 2017;28:88–100. doi:10.1016/j.scs.2016.08.026.
- [62] Afshari H, Farel R, Peng Q. Improving the Resilience of Energy Flow Exchanges in Eco-Industrial Parks: Optimization Under Uncertainty. ASCE-ASME J Risk Uncert Engrg Sys, Part B Mech Engrg 2017;3:021002. doi:10.1115/1.4035729.
- [63] Afshari H, Farel R, Gourlia J-P, Peng Q. Energy Symbioses in Eco-Industrial Parks: Models and Perspectives 2016:V004T05A022. doi:10.1115/detc2016-59965.
- [64] Timmerman J, Vandevelde L, Van Eetvelde G. Towards low carbon business park energy systems: Classification of techno-economic energy models. Energy 2014;75:68–80. doi:10.1016/j.energy.2014.05.092.
- [65] Taibi E, Gielen D, Bazilian M. The potential for renewable energy in industrial applications. Renew Sustain Energy Rev 2012;16:735–44. doi:10.1016/j.rser.2011.08.039.
- [66] Pechmann A, Zarte M. Economic Analysis of Decentralized, Electrical- and Thermal Renewable Energy Supply for Small and Medium-Sized Enterprises. Procedia CIRP 2017;61:422–7. doi:10.1016/j.procir.2016.11.143.
- [67] Kim KT, Lee DJ, Park SJ, Zhang Y, Sultanov A. Measuring the efficiency of the investment for renewable energy in Korea using data envelopment analysis. Renew Sustain Energy Rev 2015;47:694–702. doi:10.1016/j.rser.2015.03.034.
- [68] Cayir Ervural B, Zaim S, Delen D. A two-stage analytical approach to assess sustainable energy efficiency. Energy 2018;164:822–36. doi:10.1016/j.energy.2018.08.213.
- [69] Togawa T, Fujita T, Dong L, Fujii M, Ooba M. Feasibility assessment of the use of power plant-sourced waste heat for plant factory heating considering spatial configuration. J Clean Prod 2014;81:60–9. doi:10.1016/j.jclepro.2014.06.010.
- [70] Huang F, Zheng J, Baleynaud JM, Lu J. Heat recovery potentials and technologies in industrial zones. J Energy Inst 2017;90:951–61. doi:10.1016/j.joei.2016.07.012.
- [71] Marchi B, Zanoni S, Zavanella LE. Symbiosis between industrial systems, utilities and public

service facilities for boosting energy and resource efficiency. Energy Procedia 2017;128:544–50. doi:10.1016/j.egypro.2017.09.006.

- [72] Patricio J, Axelsson L, Blomé S, Rosado L. Enabling industrial symbiosis collaborations between SMEs from a regional perspective. J Clean Prod 2018;202:1120–30. doi:10.1016/j.jclepro.2018.07.230.
- [73] Zhang B, Du Z, Wang Z. Carbon reduction from sustainable consumption of waste resources: An optimal model for collaboration in an industrial symbiotic network. J Clean Prod 2018;196:821–8. doi:10.1016/j.jclepro.2018.06.135.
- [74] Sokka L, Pakarinen S, Melanen M. Industrial symbiosis contributing to more sustainable energy use - An example from the forest industry in Kymenlaakso, Finland. J Clean Prod 2011;19:285–93. doi:10.1016/j.jclepro.2009.08.014.
- [75] Ban YU, Jeong JH, Jeong SK. Assessing the performance of carbon dioxide emission reduction of commercialized eco-industrial park projects in South Korea. J Clean Prod 2016;114:124–31. doi:10.1016/j.jclepro.2015.09.083.
- [76] Shiraki H, Ashina S, Kameyama Y, Hashimoto S, Fujita T. Analysis of optimal locations for power stations and their impact on industrial symbiosis planning under transition toward lowcarbon power sector in Japan. J Clean Prod 2016;114:81–94. doi:10.1016/j.jclepro.2015.09.079.
- [77] Leung DYC, Caramanna G, Maroto-Valer MM. An overview of current status of carbon dioxide capture and storage technologies. Renew Sustain Energy Rev 2014;39:426–43. doi:10.1016/j.rser.2014.07.093.
- [78] Koytsoumpa EI, Bergins C, Kakaras E. The CO2economy: Review of CO2capture and reuse technologies. J Supercrit Fluids 2018;132:3–16. doi:10.1016/j.supflu.2017.07.029.
- [79] Guo Y, Tian J, Zang N, Gao Y, Chen L. The Role of Industrial Parks in Mitigating Greenhouse Gas Emissions from China. Environ Sci Technol 2018;52:7754–62. doi:10.1021/acs.est.8b00537.
- [80] Feng J-C, Yan J, Yu Z, Zeng X, Xu W. Case study of an industrial park toward zero carbon emission. Appl Energy 2018;209:65–78. doi:10.1016/J.APENERGY.2017.10.069.
- [81] Guo Y, Tian J, Chertow M, Chen L. Exploring Greenhouse Gas-Mitigation Strategies in Chinese Eco-Industrial Parks by Targeting Energy Infrastructure Stocks. J Ind Ecol 2018;22:106–20. doi:10.1111/jiec.12542.
- [82] Block C, Van Praet B, Windels T, Vermeulen I, Dangreau G, Overmeire A, et al. Toward a Carbon Dioxide Neutral Industrial Park: A Case Study. J Ind Ecol 2011;15:584–96. doi:10.1111/j.1530-9290.2011.00355.x.
- [83] Su S, Lu H, Zhang L, Alanne K, Yu Z. Solar energy utilization patterns for different district typologies using multi-objective optimization: A comparative study in China. Sol Energy 2017;155:246–58. doi:10.1016/j.solener.2017.06.036.

- [84] Beier J. Manufacturing Systems and Variable Renewable Electricity Supply, Springer International Publishing AG 2017; 2017. doi:10.1007/978-3-319-46639-2.
- [85] Martinez S, Michaux G, Salagnac P, Bouvier JL. Micro-combined heat and power systems (micro-CHP) based on renewable energy sources. Energy Convers Manag 2017;154:262– 85. doi:10.1016/j.enconman.2017.10.035.
- [86] Lund H, Alberg P, Connolly D, Vad B. Smart energy and smart energy systems. Energy 2017;137:556–65. doi:10.1016/j.energy.2017.05.123.
- [87] Abdmouleh Z, Gastli A, Ben-Brahim L, Haouari M, Al-Emadi NA. Review of optimization techniques applied for the integration of distributed generation from renewable energy sources. Renew Energy 2017;113:266–80. doi:10.1016/j.renene.2017.05.087.
- [88] Mallikarjun S, Lewis HF. Energy technology allocation for distributed energy resources: A strategic technology-policy framework. Energy 2014;72:783–99. doi:10.1016/j.energy.2014.05.113.
- [89] Deckmyn C, Desmet J, Van Eetvelde G. Energy management and dynamic optimisation of eco-industrial parks. Univ Power Eng Conf (UPEC), 2013 48th ... 2013:1–6.
- [90] Bacha S, Picault D, Burger B, Etxeberria-Otadui I, Martins J. Photovoltaics in microgrids: An overview of grid integration and energy management aspects. IEEE Ind Electron Mag 2015;9:33–46. doi:10.1109/MIE.2014.2366499.
- [91] Zia MF, Elbouchikhi E, Benbouzid M. Microgrids energy management systems: A critical review on methods, solutions, and prospects. Appl Energy 2018;222:1033–55. doi:10.1016/j.apenergy.2018.04.103.
- [92] Reynolds J, Ahmad MW, Rezgui Y, Hippolyte JL. Operational supply and demand optimisation of a multi-vector district energy system using artificial neural networks and a genetic algorithm. Appl Energy 2019;235:699–713. doi:10.1016/j.apenergy.2018.11.001.
- [93] Alamin YI, Álvarez JD, del Mar Castilla M, Ruano A. An Artificial Neural Network (ANN) model to predict the electric load profile for an HVAC system \*. IFAC-PapersOnLine 2018;51:26–31. doi:10.1016/j.ifacol.2018.06.231.
- [94] Sharifzadeh M, Sikinioti-Lock A, Shah N. Machine-learning methods for integrated renewable power generation: A comparative study of artificial neural networks, support vector regression, and Gaussian Process Regression. Renew Sustain Energy Rev 2019;108:513–38. doi:10.1016/j.rser.2019.03.040.
- [95] Saviozzi M, Massucco S, Silvestro F. Implementation of advanced functionalities for Distribution Management Systems: Load forecasting and modeling through Artificial Neural Networks ensembles. Electr Power Syst Res 2019;167:230–9. doi:10.1016/j.epsr.2018.10.036.
- [96] Marugán AP, Márquez FPG, Perez JMP, Ruiz-Hernández D. A survey of artificial neural network in wind energy systems. Appl Energy 2018;228:1822–36.

doi:10.1016/j.apenergy.2018.07.084.

- [97] Rodrigues E, Gomes Á, Gaspar AR, Henggeler Antunes C. Estimation of renewable energy and built environment-related variables using neural networks – A review. Renew Sustain Energy Rev 2018;94:959–88. doi:10.1016/j.rser.2018.05.060.
- [98] Jurasz J. Modeling and forecasting energy flow between national power grid and a solar– wind–pumped-hydroelectricity (PV–WT–PSH) energy source. Energy Convers Manag 2017;136:382–94. doi:10.1016/j.enconman.2017.01.032.
- [99] Mohammadi M, Noorollahi Y, Mohammadi-ivatloo B, Hosseinzadeh M, Yousefi H, Khorasani ST. Optimal management of energy hubs and smart energy hubs A review. Renew Sustain Energy Rev 2018;89:33–50. doi:10.1016/j.rser.2018.02.035.
- [100] Yang H, Xiong T, Qiu J, Qiu D, Dong ZY. Optimal operation of DES/CCHP based regional multi-energy prosumer with demand response. Appl Energy 2016;167:353–65. doi:10.1016/j.apenergy.2015.11.022.
- [101] Ma T, Wu J, Hao L, Lee WJ, Yan H, Li D. The optimal structure planning and energy management strategies of smart multi energy systems. Energy 2018;160:122–41. doi:10.1016/j.energy.2018.06.198.
- [102] Xing X, Yan Y, Zhang H, Long Y, Wang Y, Liang Y. Optimal design of distributed energy systems for industrial parks under gas shortage based on augmented ε-constraint method. J Clean Prod 2019;218:782–95. doi:10.1016/j.jclepro.2019.02.052.
- [103] Mancarella P. MES (multi-energy systems): An overview of concepts and evaluation models. Energy 2014;65:1–17. doi:10.1016/j.energy.2013.10.041.
- [104] Côté RP, Liu C. Strategies for reducing greenhouse gas emissions at an industrial park level: A case study of Debert Air Industrial Park, Nova Scotia. J Clean Prod 2016;114:352– 61. doi:10.1016/j.jclepro.2015.09.061.
- [105] Hentschel M, Ketter W, Collins J. Renewable energy cooperatives: Facilitating the energy transition at the Port of Rotterdam. Energy Policy 2018;121:61–9. doi:10.1016/j.enpol.2018.06.014.
- [106] Afshari H, Farel R, Peng Q. Challenges of value creation in Eco-Industrial Parks (EIPs): A stakeholder perspective for optimizing energy exchanges. Resour Conserv Recycl 2018;139:315–25. doi:10.1016/j.resconrec.2018.09.002.
- [107] Albino V, Fraccascia L, Savino T. Industrial Symbiosis for a Sustainable City: Technical, Economical and Organizational Issues. Proceedia Eng 2015;118:950–7. doi:10.1016/j.proeng.2015.08.536.
- [108] Yu C, Dijkema GPJ, De Jong M, Shi H. From an eco-industrial park towards an eco-city: A case study in Suzhou, China. J Clean Prod 2015;102:264–74. doi:10.1016/j.jclepro.2015.04.021.
- [109] Kim HW, Dong L, Choi AES, Fujii M, Fujita T, Park HS. Co-benefit potential of industrial and

urban symbiosis using waste heat from industrial park in Ulsan, Korea. Resour Conserv Recycl 2018;135:225–34. doi:10.1016/j.resconrec.2017.09.027.

- [110] Lund H, Werner S, Wiltshire R, Svendsen S, Thorsen JE, Hvelplund F, et al. 4th Generation District Heating (4GDH). Integrating smart thermal grids into future sustainable energy systems. Energy 2014;68:1–11. doi:10.1016/j.energy.2014.02.089.
- [111] Karner K, Theissing M, Kienberger T. Modeling of energy efficiency increase of urban areas through synergies with industries. Energy 2017;136:201–9. doi:10.1016/j.energy.2015.12.139.
- [112] Afshari H, Jaber MY, Searcy C. Extending industrial symbiosis to residential buildings: A mathematical model and case study. J Clean Prod 2018;183:370–9. doi:10.1016/j.jclepro.2018.02.148.
- [113] Le Tellier M, Berrah L, Stutz B, Audy JF, Barnabé S. Towards sustainable business parks: A literature review and a systemic model. J Clean Prod 2019;216:129–38. doi:10.1016/j.jclepro.2019.01.145.
- [114] Wang Y, Ren H, Dong L, Park HS, Zhang Y, Xu Y. Smart solutions shape for sustainable low-carbon future: A review on smart cities and industrial parks in China. Technol Forecast Soc Change 2019;144:103–17. doi:10.1016/j.techfore.2019.04.014.
- [115] Samsatli S, Samsatli NJ. A general mixed integer linear programming model for the design and operation of integrated urban energy systems. J Clean Prod 2018;191:458–79. doi:10.1016/j.jclepro.2018.04.198.
- [116] Han Y, Fan C, Xu M, Geng Z, Zhong Y. Production capacity analysis and energy saving of complex chemical processes using LSTM based on attention mechanism. Appl Therm Eng 2019. doi:10.1016/j.applthermaleng.2019.114072.
- [117] Han Y, Long C, Geng Z, Zhu Q, Zhong Y. A novel DEACM integrating affinity propagation for performance evaluation and energy optimization modeling: Application to complex petrochemical industries. Energy Convers Manag 2019. doi:10.1016/j.enconman.2018.12.120.
- [118] Carpenter J, Woodbury KA, O'Neill Z. Using change-point and Gaussian process models to create baseline energy models in industrial facilities: A comparison. Appl Energy 2018. doi:10.1016/j.apenergy.2018.01.043.
- [119] Han Y, Zeng Q, Geng Z, Zhu Q. Energy management and optimization modeling based on a novel fuzzy extreme learning machine: Case study of complex petrochemical industries. Energy Convers Manag 2018. doi:10.1016/j.enconman.2018.03.049.
- [120] He YL, Wang PJ, Zhang MQ, Zhu QX, Xu Y. A novel and effective nonlinear interpolation virtual sample generation method for enhancing energy prediction and analysis on small data problem: A case study of Ethylene industry. Energy 2018. doi:10.1016/j.energy.2018.01.059.

[121] Han Y, Wu H, Jia M, Geng Z, Zhong Y. Production capacity analysis and energy optimization of complex petrochemical industries using novel extreme learning machine integrating affinity propagation. Energy Convers Manag 2019. doi:10.1016/j.enconman.2018.11.001.